

# Board Gender Diversity and Corporate Cash Holdings

Aitzaz Ahsan Alias Sarang, Nicolas Aubert, Xavier Hollandts

# ▶ To cite this version:

Aitzaz Ahsan Alias Sarang, Nicolas Aubert, Xavier Hollandts. Board Gender Diversity and Corporate Cash Holdings. Finance, 2021, 42, pp.7-49. 10.3917/fina.421.0007 . hal-02897433

HAL Id: hal-02897433

https://hal.science/hal-02897433

Submitted on 14 Jul 2022

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers.

L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

# **Board Gender Diversity and Corporate Cash Holdings**<sup>1</sup>

Aitzaz Ahsan Alias Sarang<sup>2</sup>
Nicolas Aubert<sup>3</sup>
Xavier Hollandts<sup>4</sup>

Abstract This study investigates the relationship between boardroom gender diversity and corporate cash holdings (i.e., liquidity risk) and excess cash holdings (i.e., agency risk). Using a sample of French firms listed on the SBF 120 index over the period 1998 to 2015, we find a significant positive relation of board gender diversity for two measures of corporate cash holdings (i.e., corporate cash ratio and industry adjusted cash ratio) and a negative relation with excess cash holdings. These results support that feminized boards are associated with lower liquidity and agency risk implying that women directors engage in stricter monitoring and disallow stockpiling beyond the optimal cash level. The study provides significant intermediate results in relation to the Copé-Zimmermann gender quota law that are robust to alternate specifications and tests.

Keywords: Female Directors, Risk Propensity, Cash Holdings, Gender Quota

<sup>&</sup>lt;sup>1</sup> We are grateful to the helpful comments received from editor Christophe Godlewski, two anonymous reviewers, Dr Asad Ali Rind, the discussants and participants at various conferences and seminars including: Hubert de la

Bruslerie at the International Corporate Governance Conference (2017), Laurent Vilanova at the French Finance Association Conference (2017), participants of the EDEN EIASM Doctoral Seminar on Board and Corporate Governance participants (2019) and participants of the Aix-Marseille University doctoral workshop in Economics and Management (2017).

<sup>&</sup>lt;sup>2</sup> Aix-Marseille Univ, CERGAM, IAE Aix-Marseille, Aix-en-Provence, France & Iqra University, Karchi Pakistan; aitzaz.sarang@gmail.com

<sup>&</sup>lt;sup>3</sup> Aix-Marseille Univ, CERGAM, IAE Aix-Marseille, Aix-en-Provence, France; niaubert@gmail.com

<sup>&</sup>lt;sup>4</sup> Kedge Business School and CRCGM; xavier.hollandts@kedgebs.com

#### Introduction

A significant and growing body of literature associates board gender diversity with notable corporate outcomes. We contribute to this literature by investigating how board gender diversity affects corporate cash holdings (i.e., liquidity risk) and excess cash holdings (i.e., agency risk). An important question that has arisen in view of the greatgreat recession of 2008 is whether the situation would have been different if more women directors were represented on corporate boards. In this context, Harriet Harman, while deputy leader of the UK Labour Party, blamed the financial crisis on the male domination of corporate boards in banks (Nigel, 2009). Minow (2008) also blames boards of directors for the financial disarray. In the aftermath of the financial crisis, many countries in the EU-28 have initiated reforms that specifically set a mandatory quota for women directors on corporate boards (Kirsch, 2018). Consequently, there has been a steady, albeit incremental, increase in the presence of women on corporate boards. For example, in the EU-28, women's representation on corporate boards rose to 27.8% compared to 11.9% in 2010<sup>5</sup>, yet the economic case of board gender quota law is unclear, while others believe that the primary motive of quota laws is not an economic case but to promote equality between men and women at board levels (Bertrand et al., 2019).

With the growing participation of women directors, the decisions of corporate boards are expected to differ for at least two reasons. First, based on critical mass theory, the influence of women directors on corporate board decisions increases with their number (Torchia et al., 2011). Second, based on social role theory (Eagly, 1987), there are fundamental differences between men and women. The contemporary literature mainly investigates how board gender diversity affects corporate performance and corporate governance. However, Kirsch, (2018) discuss that

-

<sup>&</sup>lt;sup>5</sup>According to the European Union Gender statistics database: <a href="https://eige.europa.eu/gender-statistics/dgs/indicator/wmidm">https://eige.europa.eu/gender-statistics/dgs/indicator/wmidm</a> bus wmid comp compbm/datatable

the economic case of board gender quota law is unclear, while other believe that the primary motive of quota laws is not an economic case but to promote equality between men and women at board levels (Bertrand et al., 2019). Our paper addresses this gap by examining how women directors affect liquidity risk and agency risk. Although psychology and economics studies report that the average women has a lower risk appetite than the average man (Barber and Odean, 2001; Byrnes et al., 1999), whether firms with more female directors hold more cash that is, a lower liquidity risk is still unknown. We also focus on excess cash because free cash flow theory posits that managers are incentivized to hold excess cash, as this increases the assets under their control. Such excess cash is an amount that exceeds the need for day to day operations and precautionary measures, and therefore, likely to increase agency risk. The agency literature on corporate governance reports that women directors are very strict in the monitoring function and enhance board effectiveness. This increased board monitoring and board effectiveness is expected to ultimately curtails managerial discretion, and hence affect excess cash holdings. These relationships are significant as women with a seat on the board of directors are not ordinary women and, therefore, their risk behavior might differ. Further, as excessive risk-taking has been deemed a main cause of the financial crisis, board members now face heightened expectations concerning their role in risk oversight. Board gender diversity is a recent phenomenon and, thus far, no link has been established between women's representation on corporate boards and effective risk oversight.

We specifically consider corporate cash holdings and excess cash holdings as measures of liquidity risk and agency risk respectively. According to Gill and Shah (2012), corporate cash is the most liquid asset and a measure of a corporation's ability to pay its obligations on time. However, from an agency perspective, holding excess cash is also a source of agency cost since

cash leaves more discretionary power to top executives. Coinciding with the upward trend in board gender diversity, there is a worldwide positive trend in corporate cash holdings. A study by Iskandar-Datta and Jia (2012) covering seven industrialized countries finds the upward trend in cash holdings, with France showing a modest rise and Japan a significant decline. According to Moody's (2016), the cash holdings of non-financial rated firms in Europe, the Middle East, and Africa (EMEA) increased by 5% as of December 2015 compared to the previous year. Furthermore, the study shows that France, Germany, and the UK accounted for approximately 52% of total EMEA cash holdings, with France leading at 21%. These features make France an interesting case for studying corporate cash holdings. Standard financial theory refers to this upward trend of corporate cash holdings as a conundrum or a paradox, considering cash a source of agency costs (Jensen, 1986). In this regard, women directors may choose to hold more cash to avoid liquidity risk and defaulting if they are more risk-averse and experience greater anxiety in the anticipation of negative consequences. Furthermore, the questions as to what prompted the upward trend in corporate cash holdings and whether holding high levels of cash is justifiable are puzzling. Uyar and Kuzey (2014, p.1) state that cash is a "two-edged sword." The bright side is that by keeping idle cash, firms reduce their liquidity risk and the probability of financial distress; on the dark side, holding excess cash exacerbates the risk of misappropriation, invoking agency conflicts (Nikolov and Whited, 2014). In this perplexing situation, the present study also attempts to investigate how board gender diversity affects the liquidity risk measured by cash holdings and the agency risk measured by excess cash holdings.

We examine the relationship between board gender diversity on one hand and corporate cash holdings and excess cash holdings on another on a sample of the largest French listed companies from 1998 to 2015. There are specific reasons that make the French case unique and interesting

to investigate in relation to our research question. First, in 2011, French legislators passed the Copé-Zimmermann law, which mandated a quota for the representation of women on corporate boards in two phases: 20% by 2014 and 40% by 2017. Similarly, other European countries have also legally and voluntarily enacted quotas for women, but according to the (European Commission, 2016, 2012), the improvement of gender balance in France has been more drastic and swifter, the proportion of women on corporate boards having risen from 22% in 2012 to 37% in 2016. This relatively high change in board diversity in a short time makes the French case particularly interesting. Second, in France, non-financial companies distributed dividends, on average, approximately 10% of their EBITDA between 2001 and 2015 (data from Insee, the French national statistics agency). According to the Global Dividend Index published by Janus Henderson in 2016, 2017, and 2018, France is the highest dividend-paying economy in Europe.<sup>6</sup> In a country where dividend payout is so important, holding cash is more challenging. Third, to our knowledge, while the topic of board gender diversity has been examined in depth in the US, UK, Spain, and Nordic countries (Kirsch, 2018), few results have been obtained from other countries.

orporate ocial esponsibility <sup>7</sup>Investigating the French listed companies, first, our results document a significant positive relationship between board gender diversity and liquidity risk (i.e., cash ratio and industry-adjusted cash ratio) and a negative relationship with agency risk (i.e. excess cash). These results support the notion that women directors have a lower liquidity risk-taking propensity than male directors and that they engage in stricter monitoring disallowing

\_

<sup>&</sup>lt;sup>6</sup> The Global Dividend Index is published every year, and reports are downloadable at the following link: https://www.janushenderson.com/frpi/documents-finder-eu

<sup>&</sup>lt;sup>7</sup> PACTE stands for Plan d'Action pour la Croissance et la Transformation des Entreprises or Action Plan for Business Growth and Transformation. A presentation of the Law is available here: <a href="https://www.gouvernement.fr/en/pacte-the-action-plan-for-business-growth-and-transformation">https://www.gouvernement.fr/en/pacte-the-action-plan-for-business-growth-and-transformation</a>

stockpiling more than the optimal cash level. We also document that the effect of women directors is significant post gender quota law. These results are robust for several alternate specifications and tests.

The paper makes the following contributions to the literature. First, several studies have analyzed the determinants of cash holdings (Bates et al., 2009; Bigelli and Sánchez-Vidal, 2012; Atif et al., 2019) in mostly Anglo-Saxon common law countries but France is a continental European civil law country. Graff (2008) finds that investors are treated differently across legal systems. For instance, major differences exist between the Anglo-Saxon and Continental European countries. According to Rhodes and Apeldoorn (1997), in the Anglo-Saxon, ownership is diffused, the board structure is unitary, and control comes from the market. Shleifer and Vishny (1996) argue that, in Anglo-Saxon countries, the legal protection of investors is greater. Moreover, French civil law is more precarious in terms of protecting investors because it limits agency problems to a lesser degree (La Porta, 2000). Also, La Porta et al. (2008) discuss that, in comparison to French civil law, the common law countries provide greater protection to investors that leads to a developed financial system. Although France is a continental European country, Wirtz (2019) argues that French corporate governance has experienced substantial transformations since the middle 1980s. The country now shares most of the characteristics of a liberal market economy. However, the corporate governance legal differences still exist and limit the generalization of Anglo-Saxon findings over continental European civil law countries. In this context, the French case is particularly important, because France has also had a strong historical influence on the legal system in Western Europe. According to Belot et al. (2014), the legal codes of nine European countries are inherited from French civil law. Thus, the results can be better generalized to those countries (Belot et al. 2014). According to Adams and Ferreira (2009), female directors are stricter at monitoring than men, a characteristic that is important in countries where investors' legal protection is considered as low like France.

The remainder of the paper is structured as follows. Section 1 develops the French specific institutional context of the Copé-Zimmerman mandatory quotas law. Section 2 presents the literature review and hypothesis development with a focus on the relationship between board gender diversity and corporate cash holdings. Section 3 presents our empirical strategy. Section 4 reports and discusses the results and robustness analysis and their implications for future studies. Section 5 provides concluding remarks.

#### 1 The impact of mandatory quotas: The case of the Copé-Zimmermann law in France

French legislators passed the Copé-Zimmermann law in 2011. It mandated a quota for the representation of women on corporate boards. In this regard, several impacts have been highlighted in the literature. First, French boards are more gender balanced as women must make up at least 40% of directors. This effect is sometimes labeled as the "lady-boom," since many listed firms were seeking women candidates for director seats. For instance, The Financial Times (2016) reports that in 2016 women accounted for 28 of the 44 board appointments made by the CAC 40 index French firms. The article also states that the competition for appointing women directors is becoming intense and that the lack of qualified women is compelling multiple firms to appoint the same candidate. Some studies suggest that the quota laws and the fierce competition for women board members might force firms to hire directors because of their gender rather than their skills, leading to a reduction in the level of human capital on boards (Ahern and Dittmar, 2012). However, Allemand et al. (2016) report that French firms have managed to find suitable women candidates who meet the usual selection criteria for the position of director. Also, Benkraiem (2017) concludes that the Copé-Zimmermann law is more likely to

enhance the effectiveness of board decisions as these women are introduced onto boards as independent directors.

Another consequence has been to push more women to become or to be hired as executive managers, since many directors are chosen from among the corporate elite (Dang and Nguyen, 2016). The literature is sparse in the French context, but some authors underline that the presence of female directors exhibits various impacts on board dynamics, board performance, risk-taking, and financial policies (Allemand et al., 2016; Allemand and Brullebaut, 2014; Rebérioux and Roudaut, 2018). In the context of the Copé-Zimmermann law in France, Rosenblum and Daria (2015) interviewed 24 French board members, concluding that the quota laws have impacted boards' decision-making processes because some of the newly appointed women are from outside France and some are expert in a wider range of areas. The study reports that board gender diversity in France has affected the tone of board interactions. One participant noted that after her first meeting – the board's first with a woman present – the men had said to her (p.25), "That was the best meeting we ever had as we clashed about nothing and discussed things peacefully." Another participant noted that, compared to men, women tend to ask questions designed to elicit additional information. Moreover, the study also finds that women played some role in risk aversion and were more insistent on precise measurement of risk.

#### 2 Literature review and hypotheses development

Our research examines how board gender diversity affects corporate cash holdings. We present the academic literature investigating this question and then develop our hypotheses.

# 2.1 Corporate cash holdings

We investigate the association of board gender diversity with two main focuses of the academic literature on the determinants of cash holdings: liquidity risk and agency risk. Corporate cash holdings is the main way to insure corporate liquidity. In a survey on corporate liquidity management, Almeida et al. (2014; p. 135) conclude that: "While there are alternatives to cash holdings such as hedging or lines of credit, cash remains "king", in that it still is the predominate way in which firms ensure future liquidity for future investments. Traditionally, two main theoretical models explain corporate cash holdings through the determinants of capital structure (i.e., the debt to equity ratio) with the agency theory being predominant. Since increasing cash decreases net debt, any decision about cash holdings automatically affects capital structure. First, owing to information asymmetry and based on pecking order theory, Myers and Majluf (1984) conclude that firms do not maintain target cash holdings, but rely primarily on internal funds and then on external funds. Therefore, firms prefer to hold as much cash as possible. Second, contrary to pecking order theory, trade-off theory predicts that firms prefer to hold an optimal amount of cash through a trade-off between tax benefits and the bankruptcy costs associated with debt (Al-Najjar and Belghitar, 2011). Apart from these two traditional theories, free cash flow agency theory (Jensen, 1986) posits that managers hold excess cash to increase their discretionary power. Excess cash may be squandered on negative net present value projects. Under this view, holding excess cash exacerbates the risk of misappropriation resulting from agency conflicts. Empirical studies also support the propositions of this theory. Based on a sample of 45 countries, Dittmar et al. (2003) report that firms hold up to twice as much cash in countries where shareholders are not well protected by law compared to countries where they

are. Nikolov and Whited (2014) conclude that, due to agency problems, cash holdings are estimated as being 20% higher, resulting in a 6% drop in shareholder value.

#### 2.2 Board gender diversity

Since the introduction of quotas for female directors in different countries, more women have seats on the board. This increased representation of women on boards implies they are likely to have more influence on corporate decisions. Relying on critical mass theory, Konrad et al., (2008) and Torchia et al., (2011) report that women's influence on corporate board decisions increases with their number. The decisions of corporate boards are also expected to differ as more women become involved, as according to social role theory Eagly (1987), there are fundamental differences between men and women. This theory posits that women are expected to be ascribed to roles with a more communal focus, such as being helpful and kind and seeking security and safety. By contrast, men are expected to be assertive, dominant, and willing to take risks. Due to these perceived fundamental differences between men and women, and as a consequence of the growing presence of women directors on corporate boards, a large body of contemporary research primarily investigates how women affect firm performance and corporate governance.

To understand the relationship between board gender diversity and firm performance, the literature mainly focuses on agency theory (Carter et al., 2003, 2010; Terjesen et al., 2016). These studies report that women directors on corporate boards may increase board effectiveness through a monitoring and control function, resolving agency conflicts and aligning the interests of managers and shareholders (Fama and Jensen, 1983). In the agency theory context, Carter et al. (2003) argue that a more diverse board may be able to better monitor managers, as board gender diversity increases board independence. Furthermore, the authors state that, while agency

theory does not suggest a clear link between board diversity and financial performance, the possibility that board diversity is beneficial is not ruled out. Based on the agency theory rationale, Francoeur et al. (2008) report that women – like external stakeholders, ethnic minorities, or non-nationals – often bring a fresh perspective to intricate issues that may help to correct informational bias in strategy formulation and problem-solving. Relying on this theory, Dang and Vo (2012) argue that women on corporate boards increase the effectiveness of board monitoring.

Resting on this theory and other conceptual frameworks, empirical studies on the impact of board gender diversity on firm performance report conflicting results. Several studies state a positive relationship (Carter et al., 2003; Terjesen et al., 2016), providing three explanations. First, Kim and Starks (2016) conclude that women directors bring unique skills to corporate boards, which result in enhanced board advisory effectiveness. In addition, they report that women directors diversify boards' expertise more than their male counterparts. Enhanced advisory effectiveness leads to better decisions and, thus, better firm performance.

Second, studies of corporate governance report that women directors improve monitoring effectiveness (Carter et al., 2010, 2003; Erhardt et al., 2003). Compared to their male counterparts, female directors are more likely to have active functions on corporate boards (Virtanen, 2012). Empirical studies also suggest that women directors outperform men in terms of monitoring. One argument pertaining to this relates to inside directors versus outside directors. Studies report that women are more likely than men to be outside directors (Adams and Ferreira, 2009; Carter et al., 2010). Compared to inside directors, outside directors experience fewer conflicts of interest and are, therefore, more likely to play a greater role in monitoring efforts. Adams and Ferreira (2009) report that the presence of women directors is associated with

increased board meeting attendance and that they prefer to be part of monitoring committees. Furthermore, female directors face difficulties in establishing credibility and influence. To preempt this, they tend to be more thoroughly prepared for meetings (Singh et al., 2002), showing increased monitoring activity. More effective monitoring prevents agents from wasting valuable resources, thus leading to better performance. Levi et al. (2014) conclude that firms with more male directors are more likely to participate in merger and acquisition activities, paying higher acquisition premia.

Third, gender diversity on corporate boards enhances governance quality, as a heterogeneous group is more likely to address concerns than one that is homogeneous (Ely and Thomas, 2001). Other studies find a negative effect. Adams and Ferreira (2009) report that, on average, gender diversity has a negative impact on firm performance in the US. Yet, other studies report no significant effect (Erhardt et al., 2003). These inconclusive empirical results can be attributed to differences across studies in terms of performance measures, methodologies, omitted variable biases, time horizons, and other contextual issues (Adams et al., 2015).

Finally, another limited strand of literature looks at how women directors affect risk patterns. Research in economics and psychology indicates that, in general, women tend to be more risk-averse than men. A meta-analysis of 150 studies on risk-taking behavior (Byrnes et al., 1999) finds that women are less likely than men to be involved in "intellectual risk taking", "risky experiments", and "gambling." Other authors argue that women have a lower appetite than men for risk (Barber and Odean, 2001). These findings on risk aversion are consistent with the psychology literature. For instance, a first possible explanation for gender differences in risk-taking is based on interpretation of risk. Arch (1993) states that men are more likely to interpret risk as a challenging situation that requires participation, while women consider it a threat that

encourages avoidance. Another possible explanation is that emotions might explain women's risk aversion. Fujita et al. (1991) note argue that women experience greater anxiety and fear than men in anticipation of negative consequences. Therefore, if women have experienced undesirable consequences, they are more likely to become risk-averse.

Sensation seeking theory also attempts to explain gender differences in risk aversion. Sensation seeking is a preference for diverse, new, and complex sensations, typically entailing a risk factor, which may be physical, social, legal, or financial. Men score significantly higher than women for this personality trait (Zuckerman and Kuhlman, 2000), implying that this may have an effect on gender attitudes to risk aversion. Byrnes et al. (1999) also argue that sensation seeking explains risk aversion by women. Relying on this theory, the authors suggest that men would always take more risks than women and that the gap would remain relatively the same across contexts. Finally, Eagly's (1987) gender role theory may also be used to explain differences in attitudes to risk aversion between men and women. Research based on this conceptual framework shows that men and women differ psychologically in meaningful ways. The theory posits that women are ascribed more feminine characteristics, such as being helpful, kind, sympathetic, and safety and security seeking. By contrast, the male role is characterized by other qualities including a willingness to take risks (Eagly and Crowley, 1986). Research on personality traits suggests that willingness to take risks is a component of masculinity (Bem, 1974). If willingness to take risks is perceived to be part of the male role and a need for security and risk aversion is viewed as part of the female role, there is reason to expect gender differences in attitudes and behaviors associated with decisions that involve risks. Jadack et al. (1995), relying on social role theory, argue that men are more suited to risk-taking than women, as women seek safety and security and have a greater tendency to avoid harm. Applying this conceptual framework to health risk,

the authors claim that gender roles strongly explain women's greater risk aversion. Davidson and Freudenburg (1996) also argue that social roles explain women's risk aversion, and Thom (2003) reports clear evidence of gender role differences in response to risk-taking.

Women who break the glass ceiling and reach director positions may be expected to act differently to other women. To reach such positions, females in predominantly male environments may adapt their behavior so that gender differences become narrow and disappear. However, in the risk management domain, there is literature, albeit limited, that documents a negative relationship between women directors and insolvency risk (Wilson and Altanlar, 2009).

# 2.3 Hypotheses development

Based on the literature discussed above, we expect women directors to affect corporate cash holdings. Trade-off theory predicts that firms prefer holding optimal amounts of cash through a trade-off between the additional benefits and costs associated with holding cash. Studies in psychology and economics on the risk-taking behavior of women argue that they have a lower risk appetite and lower sensation seeking scores than men. Perception of gender roles also associate women with risk avoidance and, furthermore, some researchers claim that women experience more anxiety and fear than men in anticipation of negative consequences. Thus, in the trade-off between the benefits and costs of holding cash, we may expect women directors to give greater consideration to risk. Cash is the main proxy of liquidity risk (Gill and Shah, 2012) and the "king" of liquidity management tools (Almeida et al., 2014). Holding more cash can prevent firms from defaulting, and women directors may be expected to hold higher amounts of cash. Thus, our first hypothesis is:

*H1:* Corporate cash holdings are higher for firms with more gender-diverse boards.

The second argument and hypothesis is based on excess corporate cash. Agency theory posits that managers are incentivized to hold excess cash, as this increases the assets under their control. Such excess cash is an amount that exceeds the need for day to day operations and precautionary measures. However, as discussed above, agency literature and empirical studies on corporate governance report that women directors are very strict in the monitoring function and enhance board effectiveness. Board dynamic literature reports that women directors are more likely to raise questions, debate issues, and are more dogged than men in pursuing answers to difficult questions. This increased board monitoring and board effectiveness ultimately curtails managerial discretion, and hence lower excess cash holdings are expected in the presence of more women directors. Hence our second hypothesis:

*H2:* Excess cash holdings are lower in firms with more gender diversified boards.

# 3 Empirical strategy

Here, we present the data and develop our empirical strategy to deal with endogeneity issues and the model specification. First, we discuss why the relationship between corporate cash holdings and explanatory variables is endogenous and then specify the model which addresses two specific sources of endogeneity that are of high concern in board and risk studies – omitted variable and simultaneity biases. We then present the empirical model specification, including the variables. The data sources for this study are the Thomson Reuters Worldscope database and the INSEAD OEE Data Services (IODS) Corporate Governance database. Worldscope is the premier source of detailed financial statement data of public companies. We retrieved the

governance data from the IODS database, which comprehensively covers French companies listed on the SBF 120 index. Based on general industry classification we removed the financial and the utilities firms. These sectors have very specific characteristics, as their cash holdings and liquidity are governed by specific regulations and removing them is quite common. We have 27 companies in these sectors. We have 4 companies that cannot be included because of missing data. After excluding these firms, the total sample includes 89 French firms in the period between 1998 and 2015.

# 3.1 Endogeneity issues in the relationship between board gender diversity and cash holdings

Empirical research in corporate finance and corporate governance is often plagued by endogeneity issues (Adams, 2016; Jurkus et al., 2011; Roberts and Whited, 2012). The problem of endogeneity is also expected to be inherent in relation to corporate cash holdings and board gender diversity, as there is a broad consensus that board characteristics are not exogenous; rather, they are endogenously chosen to suit the firm's environment (Adams, 2016; Jurkus et al., 2011). In relation to women directors, (Jurkus et al., 2011) find that the choice to appoint female directors is not independent from characteristics of firms that board gender diversity is meant to influence. Adams, (2016) also argues that board gender diversity is not independent of other factors. The view that the appointment of women directors is a choice that firms make must be taken into consideration when estimating the gender—liquidity risk relationship. In particular, two sources of endogeneity are likely to bias our estimates: omitted variables bias and reverse causality (Adams, 2016; Sila et al., 2016). To accurately test whether women directors affect cash holdings, these two concerns must be addressed.

Regarding omitted variable bias, some variables that are difficult to observe and are not considered in various cases may affect both the liquidity risk and the appointment of a director. The inability to consider these as explanatory factors creates unobservable heterogeneity. For instance, one omitted variable in the risk equation and director appointment can be managerial ability. According to Hermalin and Weisbach, (1988), highly skilled CEOs are more effective in managing risk and, at the same time, have considerable influence over appointment of directors. The study reports that CEOs with high ability may select directors who are less likely to hold them accountable. Concerning women directors, as we have seen, the literature documents that they are more likely to be effective monitors of CEOs (Adams and Ferreira, 2009); thus, it is expected that unobserved CEO ability and CEO preferences for a less strict board might be correlated with the extent of gender diversity. However, the ability of CEOs is difficult to observe or measure. A second possibility is that, in the case of liquidity shortage, firms may appoint particular board members who have a good reputation in the financial sector - also difficult to observe or measure. The unobservable heterogeneity created by an inability to consider these factors means that these omitted variables appear in the error term. If these factors are, in fact, correlated with explanatory variables, there is an endogeneity problem that causes inference to break down.

As for reverse causality, or simultaneity bias, the observed relationship between cash and its explanatory variables, such as director appointment, growth rate, and dividend policy, may reflect the impact of the former on the latter, therefore creating a simultaneity problem. For instance, if women directors are risk-averse, they may self-select into lower-risk firms (Farrell and Hersch, 2005). A second example is that we expect an inverse relationship between corporate cash holdings and dividends as, with dividend payments, firms' distribution of cash

ultimately reduces the balance. On the other hand, corporations announce their dividend policy based on their cash position and, therefore, both are expected to impact each other. The same behavior may also be expected for capital expenditures. Thus, in either case, simultaneity is expected.

Considering the endogeneity problems, the common strategy is to identify an external instrumental variable that explains representation of women directors on the board and should be exogenous to the firm outcomes under consideration (i.e., cash holdings) (Jurkus et al., 2011). However, (Adams, 2016; Sila et al., 2016) discuss that it is challenging to find an exogenous instrumental variable for women directors. Sila et al., (2016) provide evidence that the variables employed in previous studies as a source of exogenous variation in board gender diversity are not truly exogenous. Considering these limitations, we employ a dynamic model using the generalized method of moments (GMM) techniques (Arellano and Bover, 1995; Blundell and Bond, 1998). These techniques are based on the internal instrumental variables estimation method, whereby the internally generated lagged dependent variables and explanatory regressors are used as estimators. In their studies on cash holdings (Al-Najjar and Belghitar, 2011; Bigelli and Sánchez-Vidal, 2012; Ozkan and Ozkan, 2004; Uyar and Kuzey, 2014), risk and board gender diversity (Sila et al., 2016) also employed this method.

We consider this dynamic model to be appropriate because static models implicitly assume that firms can quickly adjust to a target level of cash following random shocks or changes in firm-specific characteristics. However, cash is a dynamic variable as the adjustment process takes time because of transaction costs, causing the cash policy not to be immediately adjusted to a more desirable one. As firms set a target level of cash (Opler et al., 1999), the decisions made in previous periods explain the level achieved over time (Ozkan and Ozkan, 2004). Furthermore,

based on the insight of Wintoki et al., (2012), we argue that the reverse causality problem also has a dynamic nature – which means that the representation of women directors in the boardroom is influenced by the past risk of the firm. This is because the appointment of a director is made before the next risk realization becomes observable. Therefore, only past risk would be included in the information set considered when appointing a new female director.

#### 3.2 Model specification

The basic model specification to be tested is:

$$\begin{aligned} & \text{Corporate\_Cash}_{i,t} \ = \ \beta 0 \ + \ \beta 1 \text{CASH}_{i,t-1} \ + \ \beta 2 \text{WDIR}_{i,t} \ + \ \beta 3 \text{BIND}_{i,t} \ + \ \beta 4 \text{CEODUAL}_{i,t} \ + \\ & \beta 5 \text{BSIZE}_{i,t} \ + \ \beta 6 \text{CFO}_{i,t} \ + \ \beta 7 \text{DEBT}_{i,t} \ + \ \beta 8 \text{FSIZE}_{i,t} \ + \ \beta 9 \text{M\_B}_{i,t} \ + \ \beta 10 \text{CFO\_Vol}_{i,t} \ + \\ & \beta 11 \text{CAPEXP}_{i,t} + \beta 12 \text{DIV}_{i,t} + \beta 13 \text{Cop\_Effect} + \epsilon i,t \end{aligned}$$

The definitions of the variables included in our analyses is presented in the appendix A.

We consider corporate cash holdings as a measure of liquidity risk and excess cash as a proxy of agency costs. The liquidity risk captured by corporate cash holdings is measured using two different variables: cash ratio (CASH) and industry-adjusted cash ratio (Ind\_Adj\_Cash). Following Uyar and Kuzey (2014), we CASH is computed as cash and cash equivalent to total assets. The second proxy, Ind\_Adj\_Cash, is a measure of cash adjusted for industry effect. We consider Ind\_Adj\_Cash since, according to Harford et al. (2008), corporate cash holdings are strongly dependent on a firm's industry. Following Boubaker et al. (2015), and using Campbell's (1996) industry classification, we measure Ind\_Adj\_Cash by calculating the yearly median level of cash held by all firms within the industry. Adjusted cash is then computed as the difference between the firm's ratio of cash to net assets and the median level of this ratio for the given industry, where net assets are measured as total assets minus cash and cash equivalents.

In order to investigate how board gender diversity is related to agency costs, we measure excess cash (E\_cash) defined as the difference between actual cash and the explained value of cash predicted or explained by our econometric model. Following Belkhir, Boubaker, and Derouiche (2014); Dittmar and Mahrt-Smith (2007); Opler et al. (1999), we first measure the optimal cash that firms should hold predicted or explained by the model, and then calculate the excess cash as a difference between actual cash and our predicted value; in other words, the residual of a corporate cash level regression. The detailed methodology for measuring excess cash is described in the Appendix B.

Our main independent variable of interest is the board gender diversity. Following, Abad et al., (2017) we compute board gender diversity (WDIR) as the number of women directors divided by the total number of directors on corporate boards. We consider the alternate specifications of board gender diversity in the robustness section.

The literature underlines the need to include other board composition variables in considering the link between board gender diversity and corporate cash holdings. We include board independence, CEO duality, and board size. Board independence, referred to as BIND, is calculated as the ratio of independent board members to the total number of directors. We include CEO duality, denoted as CEODUAL, which is a dummy variable that takes the value of 1 if the CEO is also the chairperson of the board, and 0 otherwise. Third, we consider board size, referred to as BSIZE, derived from the natural logarithm of board members.

Following Boubaker et al. (2015) and Uyar and Kuzey (2014), we include a number of financial control variables. Specifically, cash flows (CFO) are earnings before interest and taxes plus depreciation and amortization, divided by total assets. We include the debt ratio (DEBT), measured as total debt divided by total assets. Then we consider the net working capital ratio

(NWC), computed as current assets minus current liabilities and cash and cash equivalents, divided by total assets. We also include firm size (FSIZE), derived from the natural logarithm of total assets. We include cash flow volatility (CFO\_Vol), measured by the standard deviation of cash flows. We also add a proxy for capital expenditure (CAPEXP), calculated as the ratio of capital expenditures to total assets. CWe also control for dividends (DIV), measured as the ratio of dividends to total assets. And finally, we consider the proxy of firm growth market to book value (M\_B), computed as the sum of the book value of liabilities and market value of equity, divided by the book value of total assets. Cop\_Effect is a dummy variable that takes the value of 1 for the year greater than 2011 inclusive, and 0 otherwise. Following Boubaker et al. (2015), all the financial variables are Winsorized at 1% and 99% to eliminate the effect of outliers.

#### 4 Results

## 4.1 Descriptive analysis

The descriptive statistics for our sampled companies from the period 1998–2015 are presented in Table 1. On average, the corporate cash holdings of French companies represent 6% of assets over the period. Among the board composition variables, the average representation of women directors on corporate boards is 13.7%, indicating that French corporate boards are dominated by male directors. The independence of corporate boards is high in France, and the average representation of independent directors remains at 47.1% over the sampled period. On average, in 58% of cases, the CEO is also the chairperson of the board. Table 1 also provides detailed information on financial variables. The mean value of cash flow is 10.9%. Furthermore, net working capital over the period is 2.7%, showing that working assets are higher than working liabilities.

#### (Insert Table 1 about here)

In Table 2, we present the results of the mean difference tests between the governance and financial variables over two interval periods, 1998–2010 and 2011–2015. We split the sample into these two intervals because in 2011 French legislators passed the Copé-Zimmermann law that required women representation on corporate boards in two phases: 20% by 2014 and 40% by 2017. Due to the implementation of the law, changes may be expected in board composition and financial variables; splitting the sample into pre- and post-implementation periods allows comparison between the two. The results in Table 2 show that there is a considerable difference in mean values at the 1% significance level for the corporate cash variable over the two intervals. Cash holdings were 5.4%, on average, over the period 1998–2010 and significantly increased to 7.6% over the period 2011–2015. Thus, on average, cash holdings increased by 2.2% after the implementation of the Copé-Zimmermann law. We also observe that board gender diversity has significantly increased as, on average, the representation of women over the period 1998–2010 was 8.5%, growing to 26.1% over the period 2011–2015. This upward trend shows that firms have adhered to the Copé-Zimmermann law. The representation of women directors on corporate boards is expected to increase further, as the law mandated 40% female representation on corporate boards by 2017.

## (Insert Table 2 about here)

Table 3 presents correlations among the variables. As shown in Table 3 the correlation among variables is mainly below 0.5, and the variance inflation factors (VIF) value measuring multicolineraity is also less than the critical value of 10 suggested by O'Brien (2007).

## (Insert Table 3 about here)

#### 4.2 Results and discussion

The results of the GMM system regression analyses are presented in Table 4. These results are based on unbalanced panel data for the period 1998–2015. The findings in columns 1 and 2 of (Table 4) show a positive effect of WDIR on CASH and Ind\_Adj\_Cash. In terms of economic significance, all else being equal, when board gender diversity increases by one standard deviation, the ratio of CASH and Ind\_Adj\_Cash increases, on average, by 0.33% and 0.46% respectively, with a standard deviation of WDIR 13.7% (Table 1).

#### (Insert Table 4 about here)

The above findings corroborate Hypothesis 1 and validate the notion that women directors contribute to reducing liquidity risk. In the risk management domain, there is literature, albeit limited, that focuses on the effect of women directors on risk (Levi et al., 2014; Sila et al., 2016; Wilson and Altanlar, 2009). These studies document conflicting results. Sila et al., 2016) find no significant effect of women director on risk. This study focuses on equity risk in the US and is based on a time horizon of 1996–2010. During this period, the representation of women directors on corporate boards was largely symbolic. Other studies report a negative relationship between women directors and risk. For example, Wilson and Altanlar (2009) find a negative relationship between women directors and insolvency risk. Levi et al. (2014) conclude that firms with more male directors are more likely to participate in riskier activities such as merger and acquisition. In this strand of the literature, we focus on alternate definitions of risk, that is, liquidity risk, and

document a negative relationship between board gender diversity and liquidity risk. Our results document a positive relationship between women directors and cash holdings, supporting the notion that women directors have a lower liquidity risk-taking propensity than male directors.

Our results on the effect of board gender diversity on liquidity risk are consistent with the broad psychology literature on women's risk aversion. For instance, according to Arch (1993), men are more likely to interpret risk as a challenging situation requiring participation, while women are more likely to consider it as a threat that encourages avoidance. The results are in line with several other studies in the economics and psychology domain (Barber and Odean, 2001; Byrnes et al., 1999), documenting that women have a lower risk appetite than men. In addition, our results are consistent with gender role theory. In the context of this theory, Jadack et al. (1995) argue that men are more suited to risk-taking than women. This theory argues that women seek safety and security and have a greater tendency to avoid harm. The authors applied this theory to health risk, claiming that gender roles strongly explain why women may be risk averse. In the present study, we extend the scope of this theory and apply it to liquidity risk, finding that women are risk averse.

Cash holdings reduce the liquidity risk but are also a two-edged sword. Indeed, excess cash can be squandered by managers on negative return projects. Agency theory posits that holding excess cash exacerbates the risk of misappropriation of these funds, invoking agency problems between shareholders and managers. Dittmar et al. (2003) document that agency costs are correlated with excess corporate cash holdings. To check how women directors, affect excess cash holdings, we included the variable excess cash in column 3 of Table 4. Our findings report a significant negative effect of board gender diversity on excess cash holdings. In terms of economic significance, all else being equal, when board gender diversity increases by one standard

deviation, the ratio of ECASH decreases, on average, by 0.84%, with a standard deviation of WDIR 13.7% (Table 1). These findings corroborate Hypothesis 2 and validate the notion that women directors are stricter at monitoring, and they tend to prevent managers from holding excess liquid resources that can be readily converted into private benefits or used for negative return projects. The results are in line with agency theory in corporate governance literature (Adams and Ferreira, 2009; Carter et al., 2003, 2010; Erhardt et al., 2003; Terjesen et al., 2016; Virtanen, 2012) associating women directors with strict and effective monitoring. Consistent with this literature, our results related to the excess cash variable measuring agency costs imply that women directors do not allow managers to hold excess cash beyond the requirements of day to day operations and precautionary measures. These results indicate that women directors not only reduce the liquidity risk but also contribute to better corporate governance by reducing the risk of the misappropriation of liquid assets on negative return projects.

We also discuss the critical mass theory effects on cash holdings. The theory posits that the influence of women directors becomes more prominent when the size of the female group reaches a critical mass or a minimum threshold (Konrad et al., 2008; Torchia et al., 2011). Relying on critical mass theory, the authors find that women's influence on corporate board decisions increases with their number because relatively high numbers of women can form a block, expand their influence and affect the decision-making culture of the board. Also, increasing the number of women directors reduces their sense of isolation and allows their voices to be heard, and they no longer report being ignored or isolated. To examine the effect of the critical mass theory, we use three dummy variable WDIR1 coded as 1 if firms have 1 women on board and 0 otherwise, WDIR2 coded as 1 if firms have two women on board and 0 otherwise, and WDIR3, dummy variable coded as 1 if firms have at least three women on board and 0 otherwise. The results are presented in Table 5.

The results show that only WDIR3 have significant and robust effect on all three dependent variables.

The results of critical mass theory are in line with the study of Konrad et al. (2008) who interviewed female CEOs, women directors, and corporate secretaries at Fortune 1000 companies. The authors quoted that, "One male CEO noticed that as more women directors were added to the board, the original female directors became more active: more willing to push their issues and were more vocal". Further, the study documents that solo woman on corporate board are often marginalized and isolated. Adding up a second woman director to corporate boards though help lessen the sense of isolation but may not always cause change. A visible shift happens when boards have three or more women directors. At this number, their voice is heard, and they don't report being ignored or isolated. At the critical mass of three, women directors are considered by other board members as directors not as "female directors". Consistently, Torchia et al. (2011) also report that one or two women directors has no significant impact on firm's innovation, while three women directors significantly increase firm's innovations.

#### (Insert Table 5 about here)

## 4.3 Robustness analysis

In this section, we perform several robustness tests to re-examine the validity of our results. First, the basic equation is rerun using different proxies of the dependent variable. Different measures of cash allow us to avoid biases of each measure. Following Gill and Shah (2012), we use other proxies of liquidity risk; namely, current ratio (CR) and cash by net assets (N\_Cash). The estimates shown in column 1 and 2 of Table 6, demonstrate that the results shown in Table 4 are

not sensitive to other proxies of liquidity risk. In column 3 of Table 6, following Boubaker, Derouiche, and Lasfer (2015) we consider the logarithmic transformation of the cash ratio, and document a positive relationship of women on cash holdings. These estimates also demonstrate that women directors have a significant negative effect on risk.

#### (Insert Table 6 about here)

In Table 7, following Abad et al. (2017), we use two indices of board gender diversity: the BLAU\_INDEX and the SHANON\_INDEX. These indices consider the number of gender categories of male and female directors, also called variety and evenness of the distribution of corporate board members, or balance. Both indices are qualitatively similar, but the SHANON\_INDEX is a logarithmic measure of diversity and, therefore, more sensitive to minor differences in corporate gender diversity. The BLAU\_INDEX value for the two categories of gender diversity ranges from 0 (when the corporate board is dominated by one category, either male or female) to a maximum of 0.5 (when both categories have equal representation). The results of indices are reported in Table 7. We find that the basic results reported in Table 4 are not sensitive to indices of board gender proxies.

#### (Insert Table 7 about here)

We also address endogeneity concerns with alternate methodologies. First, following Al-Najjar and Belghitar (2011), Ozkan and Ozkan (2004), and Uyar and Kuzey (2014), we use the GMM difference techniques developed by Arellano and Bover (1995). These techniques are based on

the internal instrumental variables estimation method, whereby the internally generated lagged dependent variables and explanatory regressors are used as estimators. In addition, the first difference of endogeneity is taken into account to eliminate the endogeneity concerns. The estimates of GMM difference in Table 8 confirm that our main results in Table 4 are not sensitive to the alternate method of controlling endogeneity.

## (Insert Table 8 about here)

It can also be argued that the significant impact of gender diversity on liquidity risk is due to firm-specific factors that may simultaneously affect both gender diversity and cash holdings. In this set-up, due to differences in firm-specific characteristics, the direct analysis of all firms might not be appropriate. To eliminate such concerns, following Sila et al. (2016), we use propensity score matching. This method serves to match gender-diverse and non-gender-diverse firms with very similar characteristics. To operationalize the propensity method, following Gull et al. (2018), we defined the treatment group as a set of firms having at least one women director and matched it with the control group of firms having no women directors but almost identical firm-specific characteristics. Bad matching occurs if the nearest neighbor is distant. In order, to decrease the probability of bad matching, following, Gull et al. (2018) we use a calliper distance of 1% without replacement<sup>8</sup>. We estimate the results using the system GMM technique to the matched sample to correct for endogeneity bias.

<sup>&</sup>lt;sup>8</sup> Matching without replacement means that the same gender-diverse firm can be matched to only one non-gender-diverse firm.

#### (Insert Table 9 about here)

Our results of matching sample in Table 9 columns 1 and 2, demonstrate a positive association between the presence of at least one woman on the board and cash holdings and a negative relation of at least one woman on board and excess cash in column 3. The estimates of propensity score demonstrate that our previous results in Table 4 are not sensitive to the alternate method of controlling endogeneity. We then also employ difference-in-difference-estimator (DID) on matched sample. DID covers the 'parallel trends' assumption; implying that two same organizations are expected to follow the same change without any treatment. Roberts and Whited (2012) discuss that if the treatment has any effect on the outcome, the effect is likely to be observed in the difference between the changes of the two firms. The DID estimator is computed as:

$$\label{eq:corporate_Cash} \begin{split} & Corporate\_Cash_{i,t} = \beta 0 \ + (WDIR1)_{i,t} + (Cop\_Effect)_{i,t} + (WDIR1 \ x \ Cop\_Effect)_{i,t} + control \\ & variables + \epsilon i,t \end{split}$$

The variable WDIR1 as defined above is a treatment group, coded as 1 if firms have at least 1 woman on board and 0 otherwise. Cop\_Effect is a post treatment period, that takes the value of 1 for post quota period and, and 0 otherwise. The results in columns 4 and 5, demonstrate a positive association between the DID and cash holdings but no significant effect on excess cash in column 6.

# 4.4 Implications for future studies

holdings, implying that women contribute to reducing firms' liquidity risk. Our results highlighting a negative relation between board gender diversity and excess cash also document that women directors are strict in monitoring and effective at resolving the agency costs of cash hoarding. Women directors not only contribute positively to corporate governance but also tend to reduce the risk of misappropriation of liquid assets. These results have several implications for stakeholders. First, diverse boards bring varied perspectives and a more balanced risk management approach. The results show that encouraging more women directors on corporate boards and giving them more voice in corporate governance and decision-making may lead to the avoidance of excessive risk-taking. The lower liquidity risk-taking behavior of women directors might justify their presence on corporate boards. Ozkan and Ozkan (2004) note that high liquidity risk increases the probability of default, but women directors significantly reduce such risk. Therefore, lower risk-taking could be one of the underlying motivations of policymakers in promoting gender diversity on corporate boards after the financial crisis. Another important implication is for corporate governance consulting firms. These advisory firms that are specialized in organizational design may encourage and advice firms to appoint women at top positions, particularly in the industries and the firms that are involved in the management of risky assets and where the capital structure is riskier. These firms may encourage the appointment of women directors because they reduce risk and they are strict at monitoring the executives and therefore, the likelihood of agency problems and squandering of firm resources are expected to be lower in the presence of diverse boards. Moreover, to benefit from board gender diversity, it is important to have a significant representation of women on

In summary, our results document a positive relationship between women directors and cash

corporate boards, as their symbolic representation shows that they do not have a significant and robust impact in small numbers. However, the results of DID suggest that the liquidity risk was taken more into account after the quota law, that is, when gender diversity became higher. The implications are consistent with the extant literature on the effect of gender quota law in France (Houanti et al., 2018; Allemand et al., 2016; Benkraiem, 2017; Rosenblum and Roithmayr, 2015). These studies underline that the presence of female directors after the implementation of the Copé-Zimmermann law exhibits various impacts on board dynamics, board performance, and other strategic policies.

The above implications of our study are not without limitations. Future research could also investigate further the concept of board diversity. The board can be diverse in terms of nationality, gender, demographic attributes, or industry experience. In this paper, we focus on gender diversity for two main reasons. First, the extant literature documents that this diversity has a broad impact on corporate governance (Harjoto et al., 2015; Terjesen et al., 2015) and risk management (Levi et al., 2014; Sila et al., 2016; Wilson and Altanlar, 2009). Second, the French Copé-Zimmermann law, required that French listed companies must exhibit at least 40% of female directors by 2017. However, as other details (e.g., nationality, demographic attributes, and industry experience) were not compulsory for listed firms, we were not able to clearly identify such variables. It would be interesting to consider these other details in future studies in a country setting where law does not mandate gender diversity on corporate boards. In addition, demographic factors may also be considered (Gull et al., 2018).

Moreover, to estimate the results we relied on archival information at the aggregate board level. This approach lessens concerns about common source bias. Keeping in mind conflicting results in the risk domain (Adams and Funk, 2011; Levi et al., 2014; Sila et al., 2016; Wilson and

Altanlar, 2009), the use of additional measures of risk based on alternate methods of data collection (e.g., analysis of board minutes, observations of board meetings, qualitative studies, and peer ratings of directors) would further strengthen our confidence in these results. Future research may also benefit from evaluating individual female director involvement from board minutes or observational studies. This could lead the research into a previously untouched, but critically significant, field of investigation. However, given the difficulty of gaining access to board members and board meeting minutes, this would be a serious challenge for corporate governance scholars. Another important limitation of our study is that we consider the sample of SBF 120 index because the targeted companies or the obligation to implement gender diversity imposed by the law applies to boards of directors of: 1) listed companies; and 2) unlisted sociétés anonymes (joint stock corporations) and sociétés en commandite par actions (limited partnerships) (i) employing an average of at least 500 people, and (ii) with revenues or total assets over 50 million euros, for the last three financial years. Since SBF companies are listed thus they fit with the context of Copé-Zimmerman law, and therefore, we consider them. These companies are large in size, have a large float and are more likely to be followed by financial analysts and rating agencies, under media limelight, etc. Therefore, the results might not be generalized to small and unlisted firms.

#### **5** Conclusion

After the financial crisis of 2008, many countries have introduced corporate governance reforms that specifically set a mandatory quota for women directors on corporate boards. Consequently, there has been an increase in the presence of women directors on these boards. However,

contemporary literature provides limited and inconsistent evidence regarding the economic effect that increased participation of women might bring to a firm.

Drawing on the largest listed French firms over an 18-year period, the present study contributes to this intense debate by examining the relationship between board gender diversity and cash holdings. We find that board gender diversity significantly increases corporate cash holdings and industry-adjusted cash. In terms of economic significance, when board gender diversity increases by one standard deviation, the ratio of corporate cash and industry-adjusted cash increases, on average, by 0.6% and 0.16% respectively. Then, we document a significant negative effect of board gender diversity on agency costs proxied by excess cash. We also document that the effect of women directors on cash holdings is significant post gender quota law.

Based on these results, we conclude that boards with a higher representation of women directors tend to reduce liquidity risk. We also conclude that women directors are stricter at monitoring as they do not allow managers to stockpile more than optimal cash. A key implication of our paper is that one of the cases for more gender diversity on corporate boards could rest upon perceptions of men being risk-prone and women being risk-averse.

#### References

- Abad, D., Lucas-Pérez, M. E., Minguez-Vera, A., & Yagüe, J. (2017). Does gender diversity on corporate boards reduce information asymmetry in equity markets? *BRQ Business Research Quarterly*, 20(3), 192–205. https://doi.org/10.1016/j.brq.2017.04.001
- Adams, R. B. (2016). Women on boards: The superheroes of tomorrow? *The Leadership Quarterly*, 27(3), 371–386. https://doi.org/10.1016/j.leaqua.2015.11.001
- Adams, R. B., de Haan, J., Terjesen, S., & van Ees, H. (2015). Board Diversity: Moving the Field Forward. *Corporate Governance: An International Review*, 23(2), 77–82. https://doi.org/10.1111/corg.12106
- Adams, R. B., & Funk, P. (2011). Beyond the Glass Ceiling: Does Gender Matter? *Management Science*, 58(2), 219–235. https://doi.org/10.1287/mnsc.1110.1452
- Adams, R., & Ferreira, D. (2009). Women in the boardroom and their impact on governance and performance. *Journal of Financial Economics*, 94(2), 291–309.
- Adams, R., & Ragunathan, V. (2017). Lehman Sisters: Retrieved from https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=3046451

- Ahern, K. R., & Dittmar, A. K. (2012). The Changing of the Boards: The Impact on Firm Valuation of Mandated Female Board Representation. *The Quarterly Journal of Economics*, 127(1), 137–197. https://doi.org/10.1093/qje/qjr049
- Allemand, I., Bédard, J., & Brullebaut, B. (2016). Efficacité des lois contraignantes et des lois souples pour promouvoir la diversité de genre dans les conseils d'administration: Une comparaison France/Canada. *Finance Contrôle Stratégie*, (19–4). Https://doi.org/10.4000/fcs.1869
- Allemand, I., & Brullebaut, B. (2014). Le capital humain des femmes récemment nommées dans les conseils d'administration des sociétés françaises cotées à Paris, Management international. Retrieved from http://www.managementinternational.ca/catalog/le-capital-humain-desfemmes-recemment-nommees-dans-les-conseils-d-administration-des-societes-françaises-cotees-a-paris.html
- Almeida, H., Campello, M., Cunha, I., & Weisbach, M. S. (2014). Corporate liquidity management: A conceptual framework and survey. Annu. Rev. Financ. Econ., 6(1), 135-162. https://doi.org/10.1146/annurev-financial-110613-034502
- Al-Najjar, B., & Belghitar, Y. (2011). Corporate cash holdings and dividend payments: Evidence from simultaneous analysis. *Managerial and Decision Economics*, 32(4), 231–241. https://doi.org/10.1002/mde.1529
- Arch, E. C. (1993). Risk-Taking: A Motivational Basis for Sex Differences. *Psychological Reports*, 73(1), 3–11. https://doi.org/10.2466/pr0.1993.73.1.3
- Arellano, M., & Bover, O. (1995). Another look at the instrumental variable estimation of error-components models. *Journal of Econometrics*, 68(1), 29–51.
- Atif, M., Liu, B., & Huang, A. (2019). Does board gender diversity affect corporate cash holdings? Journal of Business Finance & Accounting, 46(7-8), 1003-1029.
- Barber, B. M., & Odean, T. (2001). Boys will be Boys: Gender, Overconfidence, and Common Stock Investment. *The Quarterly Journal of Economics*, 116(1), 261–292. https://doi.org/10.1162/003355301556400
- Bates, T. W., Kahle, K. M., & Stulz, R. M. (2009). Why Do U.S. Firms Hold So Much More Cash than They Used To? *The Journal of Finance*, *64*(5), 1985–2021. https://doi.org/10.1111/j.1540-6261.2009.01492.x
- Belkhir, Mohamed, Sabri Boubaker, and Imen Derouiche. 2014. "Control-Ownership Wedge, Board of Directors, and the Value of Excess Cash." Economic Modelling 39:110–22.
- Belot, F., Ginglinger, E., Slovin, M. B., & Sushka, M. E. (2014). Freedom of choice between unitary and two-tier boards: An empirical analysis. *Journal of Financial Economics*, 112(3), 364–385. https://doi.org/10.1016/j.jfineco.2014.02.011
- Bem, S. L. (1974). The measurement of psychological androgyny. *Journal of Consulting and Clinical Psychology*, 42(2), 155–162. https://doi.org/10.1037/h0036215
- Benkraiem, R. (2017). Board independence, gender diversity and CEO compensation | Corporate Governance: The international journal of business in society. Retrieved from https://www.emeraldinsight.com/doi/abs/10.1108/CG-02-2017-0027
- Bertrand, M., Black, S. E., Jensen, S., & Lleras-Muney, A. (2019). Breaking the Glass Ceiling? The Effect of Board Quotas on Female Labour Market Outcomes in Norway. *The Review of Economic Studies*, 86(1), 191–239. https://doi.org/10.1093/restud/rdy032
- Bigelli, M., & Sánchez-Vidal, J. (2012). Cash holdings in private firms. *Journal of Banking & Finance*, 36(1), 26–35.

- Blundell, R., & Bond, S. (1998). Initial conditions and moment restrictions in dynamic panel data models. *Journal of Econometrics*, 87(1), 115–143.
- Boubaker, S., Derouiche, I., & Nguyen, D. K. (2015). Does the board of directors affect cash holdings? A study of French listed firms. *Journal of Management & Governance*, 19(2), 341–370. https://doi.org/10.1007/s10997-013-9261-x
- Boubaker, S., Derouiche, I., and Lasfer, M., 2015. "Geographic Location, Excess Control Rights, and Cash Holdings." International Review of Financial Analysis 42:24–37.
- Byrnes, J. P., Miller, D. C., & Schafer, W. D. (1999). Gender differences in risk taking: A meta-analysis. Psychological Bulletin, 125(3), 367–383. https://doi.org/10.1037/0033-2909.125.3.367
- Campbell, J. (1996). Understanding Risk and Return. *Journal of Political Economy*, 104(2), 298–345.
- Campbell, K., & Mínguez-Vera, A. (2008). Gender Diversity in the Boardroom and Firm Financial Performance. *Journal of Business Ethics*, 83(3), 435–451. https://doi.org/10.1007/s10551-007-9630-y
- Carter, D. A., D'Souza, F., Simkins, B. J., & Simpson, W. G. (2010). The Gender and Ethnic Diversity of US Boards and Board Committees and Firm Financial Performance. *Corporate Governance: An International Review*, 18(5), 396–414. https://doi.org/10.1111/j.1467-8683.2010.00809.x
- Carter, D. A., Simkins, B. J., & Simpson, W. G. (2003). Corporate Governance, Board Diversity, and Firm Value. *Financial Review*, *38*(1), 33–53. https://doi.org/10.1111/1540-6288.00034
- Dang, R. D., Houanti, L., Ammari, A., & Lê, N. T. (2018). Is there a 'business case' for board gender diversity within French listed SMEs. *Applied Economics Letters*, 25(14), 980–983. https://doi.org/10.1080/13504851.2017.1390308
- Dang, R., & Nguyen, D. (2016). Does Board Gender Diversity Make a Difference? New Evidence from Quantile Regression Analysis. *Management International / International Management / Gestion Internacional*, 20(2), 95–106. https://doi.org/10.7202/1046565ar
- Dang, R., & Vo, L.-C. (2012). Women on Corporate Boards of Directors: Theories, Facts and Analysis. *SpringerLink*, 3–21. https://doi.org/10.1057/9780230389304\_1
- Davidson, D. J., & Freudenburg, W. R. (1996). Gender and Environmental Risk Concerns: A Review and Analysis of Available Research. *Environment and Behavior*, 28(3), 302–339. https://doi.org/10.1177/0013916596283003
- Dittmar, A., & Mahrt-Smith, J. (2007). Corporate governance and the value of cash holdings. *Journal of Financial Economics*, 83(3), 599–634.
- Dittmar, A., Mahrt-Smith, J., & Servaes, H. (2003). International Corporate Governance and Corporate Cash Holdings. *Journal of Financial and Quantitative Analysis*, *38*(1), 111–133. https://doi.org/10.2307/4126766
- Eagly, A. H. (1987). Sex differences in social behavior: A social-role interpretation. Retrieved from https://www.scholars.northwestern.edu/en/publications/sex-differences-in-social-behavior-a-social-role-interpretation
- Eagly, A. H., & Crowley, M. (1986). Gender and helping behavior: A meta-analytic review of the social psychological literature. *Psychological Bulletin*, *100*(3), 283–308. https://doi.org/10.1037/0033-2909.100.3.283
- Ely, R. J., & Thomas, D. A. (2001). Cultural Diversity at Work: The Effects of Diversity Perspectives on Work Group Processes and Outcomes. *Administrative Science Quarterly*, 46(2), 229–273. https://doi.org/10.2307/2667087

- Erhardt, N. L., Werbel, J. D., & Shrader, C. B. (2003). Board of Director Diversity and Firm Financial Performance. *Corporate Governance: An International Review*, 11(2), 102–111. https://doi.org/10.1111/1467-8683.00011
- European Commission. (2016). *Gender balance on corporate boards Europe is cracking the glass ceiling*. Retrieved from http://ec.europa.eu/justice/gender-equality/files/gender\_balance\_decision\_making/1607\_factsheet\_final\_wob\_data\_en.pdf
- European Commission, (2012). Women in economic decision-making in the EU: Progress report.
- Fama, E. F., & Jensen, M. C. (1983). Separation of Ownership and Control. *The Journal of Law & Economics*, 26(2), 301–325.
- Farrell, K. A., & Hersch, P. L. (2005). Additions to corporate boards: The effect of gender. *Journal of Corporate Finance*, 11(1), 85–106. https://doi.org/10.1016/j.jcorpfin.2003.12.001
- Ferreira, D. (2015). Board Diversity: Should We Trust Research to Inform Policy? *Corporate Governance: An International Review*, 23(2), 108–111. https://doi.org/10.1111/corg.12092
- Francoeur, C., Labelle, R., & Sinclair-Desgagné, B. (2008). Gender Diversity in Corporate Governance and Top Management. *Journal of Business Ethics*, 81(1), 83–95. https://doi.org/10.1007/s10551-007-9482-5
- Fujita, F., Diener, E., & Sandvik, E. (1991). Gender differences in negative affect and well-being: The case for emotional intensity. *Journal of Personality and Social Psychology*, 61(3), 427–434. https://doi.org/10.1037/0022-3514.61.3.427
- García-Teruel, P. J., & Martínez-Solano, P. (2008). On the Determinants of SME Cash Holdings: Evidence from Spain. *Journal of Business Finance & Accounting*, 35(1–2), 127–149. https://doi.org/10.1111/j.1468-5957.2007.02022.x
- Gill, A., & Shah, C. (2012). Determinants of Corporate Cash Holdings: Evidence from Canada. *International Journal of Economics and Finance*, 4(1), 70.
- Graff, Michael. 2008. "Law and Finance: Common Law and Civil Law Countries Compared—An Empirical Critique." Economica 75(297):60–83.
- Gull, Ammar Ali, Mehdi Nekhili, Haithem Nagati, and Tawhid Chtioui. 2018. "Beyond Gender Diversity: How Specific Attributes of Female Directors Affect Earnings Management." The British Accounting Review 50(3):255–74.
- Gustafsod, P. E. (1998). Gender Differences in Risk Perception: Theoretical and Methodological erspectives. *Risk Analysis*, *18*(6), 805–811. https://doi.org/10.1111/j.1539-6924.1998.tb01123.x
- Harford, J., Mansi, S. A., & Maxwell, W. F. (2008). Corporate governance and firm cash holdings in the US. *Journal of Financial Economics*, 87(3), 535–555.
- Harjoto, M., Laksmana, I., & Lee, R. (2015). Board Diversity and Corporate Social Responsibility. *Journal of Business Ethics*, *132*(4), 641–660. https://doi.org/10.1007/s10551-014-2343-0
- Hermalin, B., & Weisbach, M. (1988). The Determinants of Board Composition. *RAND Journal of Economics*, 19(4), 589–606.
- Hinz, R. P., McCarthy, D. D., & Turner, J. A. (1997). Are Women Conservative Investors? Gender Differences in Participant Directed Pension Investments Retrieved from website: http://econpapers.repec.org/paper/woppennpr/96-17.htm
- Iskandar-Datta, M. E., & Jia, Y. (2012). Cross-country analysis of secular cash trends. *Journal of Banking & Finance*, *36*(3), 898–912. https://doi.org/10.1016/j.jbankfin.2011.10.003

- Jadack, R. A., Hyde, J. S., & Keller, M. L. (1995). Gender and knowledge about HIV, risky sexual behavior, and safer sex practices. *Research in Nursing & Health*, 18(4), 313–324. https://doi.org/10.1002/nur.4770180405
- Jensen, M. C. (1986). Agency Costs of Free Cash Flow, Corporate Finance, and Takeovers. *The American Economic Review*, 76(2), 323–329.
- Jurkus, A. F., Park, J. C., & Woodard, L. S. (2011). Women in top management and agency costs. *Journal of Business Research*, 64(2), 180–186. https://doi.org/10.1016/j.jbusres.2009.12.010
- Kim, D., & Starks, L. T. (2016). Gender Diversity on Corporate Boards: Do Women Contribute Unique Skills? *The American Economic Review*, 106(5), 267–271. https://doi.org/10.1257/aer.p20161032
- Kirsch, A. (2018). The gender composition of corporate boards: A review and research agenda. *The Leadership Quarterly*, 29(2), 346–364. https://doi.org/10.1016/j.leaqua.2017.06.001
- Konrad, A. M., Kramer, V., & Erkut, S. (2008). Critical Mass: The Impact of Three or More Women on Corporate Boards. *Organizational Dynamics*, *37*(2), 145–164. https://doi.org/10.1016/j.orgdyn.2008.02.005
- La Porta, Rafael, Florencio Lopez-de-Silanes, and Andrei Shleifer. 2008. "The Economic Consequences of Legal Origins." Journal of Economic Literature 46(2):285–332.
- La Porta, 2000. "Agency Problems and Dividend Policies around the World.", The Journal of Finance." (https://onlinelibrary.wiley.com/doi/abs/10.1111/0022-1082.00199)Levi, M., Li, K., & Zhang, F. (2014). Director gender and mergers and acquisitions. Journal of Corporate Finance, 28(Supplement C), 185–200. https://doi.org/10.1016/j.jcorpfin.2013.11.005
- Minow, N. (2008). Commentary: Blame boards of directors for financial mess. CNN.com. Retrieved from http://edition.cnn.com/2008/POLITICS/09/18/minow.pay/
- Myers, S. C., & Majluf, N. S. (1984). Corporate financing and investment decisions when firms have information that investors do not have. *Journal of Financial Economics*, *13*(2), 187–221. https://doi.org/10.1016/0304-405X(84)90023-0
- Nigel, M. (2009). Harriet Harman: "If only it had been Lehman Sisters." Retrieved from The Independent website: http://www.independent.co.uk/news/uk/home-news/harriet-harman-if-only-it-had-been-lehman-sisters-1766932.html
- Nikolov, B., & Whited, T. M. (2014). Agency Conflicts and Cash: Estimates from a Dynamic Model. *The Journal of Finance*, 69(5), 1883–1921. https://doi.org/10.1111/jofi.12183
- O'brien, R. M. (2007). A Caution Regarding Rules of Thumb for Variance Inflation Factors. *Quality & Quantity*, 41(5), 673–690. https://doi.org/10.1007/s11135-006-9018-6
- Opler, T., Pinkowitz, L., Stulz, R., & Williamson, R. (1999). The determinants and implications of corporate cash holdings. *Journal of Financial Economics*, 52(1), 3–46. https://doi.org/10.1016/S0304-405X(99)00003-3
- Ozkan, A., & Ozkan, N. (2004). Corporate cash holdings: An empirical investigation of UK companies. *Journal of Banking & Finance*, 28(9), 2103–2134.
- Rebérioux, A., & Roudaut, G. (2018). Le Conseil D'administration : Enjeux de Gouvernance et de Responsabilité, Revue d'Économie Financière.
- Rhodes, Martin, and Bastiaan van Apeldoorn. 1997. "Capitalism versus Capitalism in Western Europe." Pp. 171–89 in Developments in West European Politics, edited by M. Rhodes, P. Heywood, and V. Wright. Macmillan Education UK.
- Roberts, M. R., & Whited, T. M. (2012). *Endogeneity in Empirical Corporate Finance*: Retrieved from https://papers.ssrn.com/abstract=1748604

- Rosenblum, D., & Daria, R. (2015). More Than A Woman: Insights into Corporate Governance After the French Sex Quota. *INDIANA LAW REVIEW*, 48, 43.
- Rosenblum, D., & Roithmayr, D. (2015). More Than a Woman: Insights into Corporate Governance After The French Sex Quota. 2015.
- Shleifer, Andrei, and Robert W. Vishny. 1996. *A Survey of Corporate Governance. Working Paper*. 5554. National Bureau of Economic Research.
- Sila, V., Gonzalez, A., & Hagendorff, J. (2016). Women on board: Does boardroom gender diversity affect firm risk? *Journal of Corporate Finance*, *36*, 26–53. https://doi.org/10.1016/j.jcorpfin.2015.10.003
- Singh, V., Kumra, S., & Vinnicombe, S. (2002). Gender and Impression Management: Playing the Promotion Game. *Journal of Business Ethics*, *37*(1), 77–89. https://doi.org/10.1023/A:1014782118902
- Terjesen, S., Aguilera, R. V., & Lorenz, R. (2015). Legislating a Woman's Seat on the Board: Institutional Factors Driving Gender Quotas for Boards of Directors. *Journal of Business Ethics*, 128(2), 233–251. https://doi.org/10.1007/s10551-014-2083-1
- Terjesen, S., Couto, E. B., & Francisco, P. M. (2016). Does the presence of independent and female directors impact firm performance? A multi-country study of board diversity. *Journal of Management & Governance*, 20(3), 447–483.
- Thom, B. (2003). Risk-taking behaviour in men Substance use and gender.
- Times, F. (2016). French boards rush to meet quotas for female membership.
- Torchia, M., Calabrò, A., & Huse, M. (2011). Women Directors on Corporate Boards: From Tokenism to Critical Mass. *Journal of Business Ethics*, 102(2), 299–317. https://doi.org/10.1007/s10551-011-0815-z
- Uyar, A., & Kuzey, C. (2014). Determinants of corporate cash holdings: Evidence from the emerging market of Turkey. *Applied Economics*, 46(9), 1035–1048. https://doi.org/10.1080/00036846.2013.866203
- Virtanen, A. (2012). Women on the boards of listed companies: Evidence from Finland. *Journal of Management & Governance*, 16(4), 571–593.
- Wall Street Journal. (2019). U.S. Corporate Cash Piles Drop to Three-Year Low. *Wall Street Journal*. Retrieved from https://www.wsj.com/articles/u-s-corporate-cash-piles-drop-to-three-year-low-11560164400
- Wilson, N., & Altanlar, A. (2009). *Director Characteristics, Gender Balance and Insolvency Risk: An Empirical Study* https://papers.ssrn.com/abstract=1932107
- Wintoki, M. B., Linck, J. S., & Netter, J. M. (2012). Endogeneity and the dynamics of internal corporate governance. *Journal of Financial Economics*, 105(3), 581–606. https://doi.org/10.1016/j.jfineco.2012.03.005
- Wirtz, Peter. 2019. "Changing Corporate Governance in France in the Late Twentieth Century." Pp. 1–21 in The Palgrave Handbook of Management History. Cham: Springer International Publishing.
- Zuckerman, M., & Kuhlman, D. M. (2000). Personality and Risk- Taking: Common Bisocial Factors. *Journal of Personality*, 68(6), 999–1029. https://doi.org/10.1111/1467-6494.00124

Table 1 Descriptive analysis of the variables included in the baseline model

Variable	Mean	Standard Deviation	Minimum	Maximum
CASH	0.0608	0.0528	0.0019	0.3252
WDIR	0.1370	0.1327	0.0000	0.6666
BIND	0.4710	0.2083	0.0000	1.0000
CEODUAL	0.5808	0.4935	0.0000	1.0000
BSIZE	11.180	3.6260	3.0000	22.000
CAPEXP	4.7189	3.9863	0.2300	25.700
FSIZE	15.253	1.6164	11.419	18.224
DIV	0.0152	0.0161	0.0000	0.0833
CFO	0.1094	0.0655	-0.0973	0.3472
DEBT	0.2457	0.1453	0.0024	0.6917
CFO_Vol	0.0199	0.0274	0.0002	0.1743
M_B	2.4730	2.7952	0.2300	36.800

Definitions of the variables are explained in Table A.2 Table 2 Univariate analysis: difference of mean test

Variable	Mean 1998 – 2010	Mean 2011 – 2015	Mean Difference	T-Test Equal Variance
CASH	0.054	0.076	0.022	7.61***
WDIR	0.085	0.261	0.175	29.2***
BIND	0.447	0.515	0.068	5.57***
CEODUAL	0.576	0.590	0.015	0.55
BSIZE	10.90	11.85	-0.953	4.67***
CAPEXP	5.130	3.640	-1.488	-6.71***
FSIZE	15.10	15.60	0.500	6.08***
DIV	0.014	0.016	0.002	2.63***
CFO	0.113	0.097	-0.015	4.11***
DEBT	0.248	0.239	-0.008	-1.00
M_B	2.722	1.882	-0.839	-5.35***
CFO_Vol	0.021	0.016	-0.004	-2.77***

<sup>\*\*\*</sup> p<0.01, \*\* p<0.05, \* p<0.1.

Table 3 Correlation analysis and variance inflation factors

Variable	CASH	WDIR	BIND	CEODUA L	BSIZE	CFO	M_B	D T	DEB	CAPEX P		DI V			CFO_Vo	VI F
CASH	1															
WDIR	0.0947	1													0 8	
BIND	0.1082	0.0237	1												1 1 9	
CEODUA L	0.0167	0.0362	0.2147	1											1 0 9	
BSIZE	0.0803	0.0642	0.1645	-0.0226	1										2 0 1	
CFO	0.1166	0.0432	0.0092	-0.0040	0.1061	1									1 4 5	
M_B	0.0630	0.0154	0.1711	0.0213	0.1250	0.2648	1								1 1 7	
DEBT	0.0608	0.0502	0.0731	0.1017	0.0328	0.2159	0.0801		1						1. 14	
CAPEXP	0.1063	0.1383	0.0967	0.0200	0.0414	0.2502	0.1462	0.0708		1					1. 15	
FSIZE	0.1136	0.0300	0.2628	-0.1109	0.6651	- 0.1947	0.0800	0.1248	0.036		1				2. 09	
DIV		0.0958	0.0582	-0.0907	0.1384	0.4181	0.2028	-0.2835	0.007			1			1. 42	
CFO_Vol	0.1309	_	0.0077	0.0363	0.1865	0.0966	0.0350	0.0376	0.066	-0.2037	- 0.0995		1	1.10		

Mean VIF

Table 4 The relationship between board gender diversity and corporate cash holdings (1998–2015), GMM system

	(1)	(2)	(3)	
	CASH		Ind_Adj	
	CASH		ma_/ taj	
L.CASH	0.4950***			
	(0.0179)			
L.Ind_Adj			0.4520***	
			(0.0370)	
L.ECASH			0.5300***	
			(0.0222)	
WDIR				
	0.0247***	0.0341***		
	(0.0070)	(0,000.4)	(0.0126)	
DIND	(0.0072)	(0.0094)	(0.0136)	
BIND				
		-0.0133**	-0.0081	
	(0.0052)	(0.0109)	(0.0069)	
CEODUAL	0.0010	-0.0003	0.0076**	
CLODETIL	(0.0010)	(0.0039)	(0.0015)	
BSIZE	-0.0003	0.0304***	0.0047	
BSIEE	(0.0026)	(0.0092)	(0.0056)	
CFO	0.0548***	0.0363	0.0615***	
CI O			(0.0147)	
FSIZE	(0.0149) -0.0014*	(0.0278) -0.0123**	0.0147)	
· S.E.L	(0.0008)	(0.0055)	(0.0019)	
DIV	-0.1570**	-0.2250*	-0.0668	
<u></u> 1	(0.0786)	(0.1290)	(0.1370)	
CAPEXP	-0.0006***	-0.0005***	0.0003***	
C. II 1211	(0.0000)	(0.0003)	(0.0000)	
M_B	-0.0001**	0.0001	-0.0002	
м_р	-0.0001 · ·	0.0001	-0.0002	

		(0.0000)	(0.0003)	( 0 0 0 0 4 )
CFO_Vol	0.1430***	0.1460**	0.0633***	,
	(0.0125)	(0.0624)	(0.0164)	
DEBT	0.0149	-0.0225	-0.0056	
	(0.0108)	(0.0164)	(0.0118)	
Cop_Effect	0.0111***	0.0019	0.0002	
	(0.0010)	(0.0014)	(0.0014)	
Constant	0.0456***	0.1290	-0.2410***	
	(0.0123)	(0.0799)	(0.0200)	
Observations	1,145	1,057	866	
AR(1)	0.0000	0.0000	0.0000	
AR(2)	0.0731	0.1456	0.1647	
Sargan Test	1	1	1	

Standard errors are in parentheses \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

Table 5 The relationship between board gender diversity and corporate cash holdings (The critical mass WDIR1, WDIR2, and WDIR 3), GMM System

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	CASH	Ind_Adj	ECASH	CASH	Ind_Adj	ECASH	CASH	Ind_Adj	ECASH
L.CASH	0.5170***			0.5220***			0.4780***		
	(0.0228)			(0.0233)			(0.0305)		
L.Ind_Adj		0.4340***			0.4110***			0.3980***	
		(0.0370)			(0.0423)			(0.0397)	
L.ECASH			0.5200***			0.5180***			0.5500***
WID ID 4	0.0000	0.004	(0.0348)			(0.0431)			(0.0227)
WDIR1	0.0029	-0.0061	-0.0081**						
WDID2	(0.0042)	(0.0040)	(0.0040)	0.0065**	0.0011	-0.0105**			
WDIR2									
WDIR3				(0.0028)	(0.0017)	(0.0047)	0.0092***	0.0046*	-0.0090**
WDIKS							(0.0032)	(0.0025)	(0.0036)
BIND	-0.0148*	-0.0148	-0.0004	-0.0247***	-0.0299*	-0.0107	-0.0053	-0.0121	-0.0053
BIND	(0.0084)	(0.0146)	(0.0179)	(0.0082)	(0.0157)	(0.0162)	(0.0090)	(0.0110)	(0.0067)
CFO	0.0577***	0.1120**	0.0823***	0.0684***	0.0732*	0.0221	0.1090***	0.0668**	0.0571***
Cro	(0.0180)	(0.0473)	(0.0318)	(0.0196)	(0.0374)	(0.0260)	(0.0186)	(0.0339)	(0.0152)
M_B	0.0001	-0.0002	-0.0004	0.0002*	-0.0005	-0.0007*	0.0003***	-0.0003	-0.0000
M_D	(0.0001)	(0.0004)	(0.0003)	(0.0002)	(0.0003)	(0.0003)	(0.0003)	(0.0003)	(0.0004)
DEBT	0.0066	0.0004)	0.0057	0.0257**	-0.0062	-0.0171	0.0227	0.0004)	-0.0034
DEDI									
EGIZE	(0.0120)	(0.0201)	(0.0206)	(0.0127)	(0.0208)	(0.0192)	(0.0147)	(0.0205)	(0.0087)
FSIZE	-0.0025	0.0008	0.0147***	-0.00134	-0.0010	0.0071	-0.0007	-0.0052	0.0154***
CAPEVE	(0.0029)	(0.0051)	(0.0031)	(0.0030)	(0.0064)	(0.0054)	(0.0025)	(0.0056)	(0.0021)
CAPEXP	-0.0003***	-0.0003***	0.0006***	-0.0004***	-0.0004***	0.0004***	-0.0004***	-0.0006***	0.0003***
CEO II I	(0.0000)	(0.0001)	(0.0001)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)
CFO_Vol	0.1640***	0.1200**	0.0424	0.0854*	0.1320**	0.0517	0.1350***	0.1970***	0.0724***
GEODINA	(0.0423)	(0.0543)	(0.0429)	(0.0473)	(0.0564)	(0.0365)	(0.0503)	(0.0572)	(0.0186)
CEODUAL	-0.0025	0.0015	-0.0011	0.0000	-0.0018	0.0068	-0.0048	0.0024	0.0049***
	(0.0027)	(0.0053)	(0.0067)	(0.00267)	(0.0031)	(0.0061)	(0.0033)	(0.0038)	(0.0013)
BSIZE	0.0061	0.0229***	0.0150	-0.0021	0.0023	0.0147**	0.0105	0.0172**	0.0074
	(0.0045)	(0.0086)	(0.0093)	(0.0073)	(0.0084)	(0.0071)	(0.0066)	(0.0076)	(0.0058)
DIV	-0.0151	-0.2830**	-0.5190***	-0.0745	-0.2610*	-0.2370**	-0.1340	-0.2760*	-0.3170***
	(0.0914)	(0.1440)	(0.1110)	(0.1050)	(0.1440)	(0.1120)	(0.0975)	(0.1440)	(0.1220)
Cop_Effect	0.0134***	0.0049***	-0.0053***	0.0117***	0.0069***	-0.0019	0.0094***	0.0032**	-0.0031*
	(0.0012)	(0.0014)	(0.0018)	(0.0014)	(0.0012)	(0.0022)	(0.0017)	(0.0015)	(0.0017)
Constant	0.0475	-0.0587	-0.2620***	0.0478	0.0291	-0.1360*	0.0028	0.0455	-0.2590***
	(0.0397)	(0.0714)	(0.0459)	(0.0548)	(0.0933)	(0.0776)	(0.0343)	(0.0832)	(0.0208)
Observations	1,145	1,057	866	1,145	1,057	866	1,145	1,057	866
AR(1)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
AR(2)	0.0728	0.1520	0.1575	0.0658	0.1315	0.1117	0.0604	0.1315	0.1687
Standard error	1	1 theses *** n<	1	1	1	1	1	1	1

Standard errors are in parentheses \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

Table 6 Sensitivity analysis for cash proxy (1998- 2015); GMM System

	(1)	(2)	(3)
	CR	NCASH	LNCASH
L.CR	0.4500***		
	(0.0218)		
L.NCASH		0.4940***	
		(0.0124)	
L.LNCASH			0.4780***
			(0.0268)
WDIR	0.0882**	0.0381***	0.2580**
	(0.0447)	(0.0071)	(0.1310)
BIND	-0.1950***	-0.0229***	-0.0354
	(0.0407)	(0.0044)	(0.0967)
BSIZE	0.0993***	-0.0049	0.0074
	(0.0378)	(0.0031)	(0.0777)
CEODUAL	-0.0257	-0.0000	0.0551***
	(0.0171)	(0.0013)	(0.0209)
CFO	0.8700***	0.0762***	0.9290***
	(0.0996)	(0.0106)	(0.2620)
M_B	-0.0047***	-0.0004***	0.0144***
	(0.0008)	(0.0000)	(0.0016)
DEBT	-0.1840***	0.0203**	0.1320
	(0.0607)	(0.0101)	(0.2840)
CAPEXP	-0.0064***	-0.0006***	-0.0103***
	(0.0005)	(0.0000)	(0.0017)
FSIZE	-0.0312***	-0.0007	-0.0289*
	(0.0110)	(0.0011)	(0.0157)
DIV	-0.4620	-0.1710*	-8.1650***
	(0.5160)	(0.0901)	(1.6990)
CFO_Vol	-0.0047	0.1970***	2.2430***
	(0.1060)	(0.0188)	(0.8580)
Cop_Effect	0.0201***	0.0139***	0.2030***
	(0.0064)	(0.0009)	(0.0224)
Constant	1.0370***	0.0481***	-1.2970***
	(0.1620)	(0.0156)	(0.281)
Observations	1,151	1,145	1,145
AR(1)	0.0000	0.0000	0.0000
AR(2)	0.4816	0.0976	0.0696
Sargan Test	1	1	1

Definitions of the variables are explained in Table A.2. Standard errors are in parentheses \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

Table 7 Sensitivity analysis using alternate measure of board gender diversity Blau and Shanon Index (1998- 2015); GMM System

	(1)	(2)	(3)	(4)	(5)	(6)
	CASH	Ind_Adj	<b>ECASH</b>	CASH	Ind_Adj	<b>ECASH</b>
L.CASH	0.4810***			0.4970***		
	(0.0173)			(0.0162)		
L.Ind_Adj		0.4410***			0.4590***	
		(0.0159)			(0.0421)	
L.ECASH			0.5440***			0.5430***
			(0.0187)			(0.0185)
BLAU_INDEX	0.0169***	0.0159**	-0.0307***			
	(0.0055)	(0.0074)	(0.0080)			
SHANON_INDEX				0.0121***	0.0111**	-0.0237***
				(0.0042)	(0.0048)	(0.0062)
BIND	-0.0150***	-0.0218***	0.0038	-0.0199***	-0.0277***	0.0045
	(0.0055)	(0.0062)	(0.0059)	(0.0056)	(0.0082)	(0.0058)
CEODUAL	0.0010	0.0035**	0.0042**	0.0013	0.0026	0.0044**
	(0.0010)	(0.0014)	(0.0018)	(0.0012)	(0.0039)	(0.0018)
BSIZE	-0.0028	0.0061	-0.0011	0.0004	0.0205***	-0.0015
	(0.0033)	(0.0044)	(0.0057)	(0.0023)	(0.0075)	(0.0057)
CFO	0.0487***	0.0704***	0.0583***	0.0502***	0.0450	0.0579***
	(0.0144)	(0.0206)	(0.0161)	(0.0142)	(0.0429)	(0.0160)
M_B	-0.0001**	0.0012***	0.0001	-0.0002**	0.0002	0.0001
	(0.0000)	(0.0003)	(0.0003)	(0.0000)	(0.0003)	(0.0003)
CAPEXP	-0.0005***	-0.0010***	0.0001***	-0.0005***	-0.0006***	0.0002***
	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)
FSIZE	-0.0006	-0.0010	0.0157***	-0.0009	0.0000	0.0157***
	(0.0010)	(0.0010)	(0.0016)	(0.0008)	(0.0041)	(0.0016)
DIV	-0.1870**	-0.3650***	-0.1880	-0.1280	-0.3470**	-0.1870
	(0.0815)	(0.0932)	(0.1340)	(0.0830)	(0.1440)	(0.1330)
CFO_Vol	0.1540***	0.1940***	0.0543***	0.1640***	0.1860***	0.0527**
	(0.0115)	(0.0227)	(0.0208)	(0.0163)	(0.0665)	(0.0207)
DEBT	0.0141	0.0166	-0.0015	0.0201*	-0.0062	0.0002
	(0.0121)	(0.0199)	(0.0121)	(0.0103)	(0.0247)	(0.0122)
Cop_Effect	0.0118***	0.0038***	-0.0015	0.0122***	0.0020*	-0.0013
	(0.0009)	(0.0011)	(0.0014)	(0.0011)	(0.0011)	(0.0015)
Constant	0.0420***	0.0044	-0.2440***	0.0362***	-0.0322	-0.2450***
	(0.0122)	(0.0116)	(0.0159)	(0.0135)	(0.0605)	(0.0158)
Observations	1,145	1,057	866	1,145	1,057	866
AR(1)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
AR(2)	0.0706	0.1532	0.1780	0.0756	0.1751	0.1597
Sargan Test	1	1	1	1	1	1

Definitions of the variables are explained in Table A.2. Standard errors are in parentheses \*\*\* p<0.01, \*\* p<0.05, \* p<0.

Table 8 Dynamic panel data GMM difference estimation results

	(1)	(2)	(3)
	CASH	Ind_Adj	ECASH
L.CASH	0.4340***		
	(0.0234)		
L.Ind_Adj		0.3940***	
		(0.0395)	
L.ECASH			0.1990***
			(0.0302)
WDIR	0.0173**	0.0350***	-0.0346**
	(0.0084)	(0.0073)	(0.0162)
BIND	-0.0056	-0.0031	0.0000
	(0.0080)	(0.0085)	(0.0090)
CEODUAL	-0.0021	-0.0025	0.0072***
	(0.0027)	(0.0045)	(0.0017)
BSIZE	0.0106*	0.0216**	0.0081
	(0.0063)	(0.0100)	(0.0070)
CFO	-0.0011	0.0024	0.0991***
	(0.0157)	(0.0355)	(0.0240)
M_B	0.0003***	-0.0000	-0.0010**
	(0.0000)	(0.0002)	(0.0004)
CAPEXP	-0.0004***	-0.0003***	0.0011***
	(0.0000)	(0.0000)	(0.0000)
FSIZE	-0.0014	-0.0045	0.0020
	(0.0037)	(0.0039)	(0.0021)
DIV	-0.0288	-0.1280	-0.0873
	(0.1190)	(0.1060)	(0.1020)
CFO_Vol	0.0386	0.0996**	0.0580**
	(0.0406)	(0.0438)	(0.0226)
DEBT	-0.0130	-0.0115	0.0325*
	(0.0149)	(0.0151)	(0.0169)
Cop_Effect	0.0079***	-0.0031**	-0.0022
	(0.0012)	(0.0015)	(0.0016)
Constant	0.0358	0.0308	
	(0.0543)	(0.0622)	
Observations	1,035	953	736
AR (1)	0.0000	0.0000	0.0000
AR (2)	0.0504	0.1268	0.0692
Sargan Test	1	1	1

Definitions of the variables are explained in Table A.2. Standard errors are in parentheses \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Table:9 Effect of board gender diversity on corporate cash, Propensity Score Matching and Difference in Difference

	(1)	(2)	(3)	(4)	(5)	(6)
	CASH	Ind_Adj	<b>ECASH</b>	CASH	Ind_Adj	<b>ECASH</b>
L.CASH	0.4580***			0.4790***		
	(0.0107)			(0.0236)		
L.Ind_Adj		0.3750***			0.3340***	
·		(0.0194)			(0.0220)	
L.ECASH		,	0.5640***		` ,	0.5840***
			(0.0343)			(0.0304)
WDIR1	0.0035***	0.0056**	-0.0099***	-0.0015	0.0058*	-0.0051**
	(0.0009)	(0.0025)	(0.0027)	(0.0011)	(0.0030)	(0.0022)
BIND	-0.0189***	-0.0215***	-0.0096	-0.0169***	-0.0173**	-0.0050
	(0.0052)	(0.0071)	(0.0078)	(0.0050)	(0.0069)	(0.0126)
CEODUAL	-0.0020**	-0.0041	0.0002	-0.0024**	0.0017	-0.0015
	(0.0008)	(0.0033)	(0.0040)	(0.0011)	(0.0057)	(0.0040)
BSIZE	-0.0023	0.0046	0.0026	0.0002	0.0057	-0.0059
	(0.0016)	(0.0040)	(0.0052)	(0.0033)	(0.0043)	(0.0064)
CFO	0.1060***	0.0593***	0.1000***	0.1300***	0.0566***	0.0776***
	(0.0085)	(0.0171)	(0.0148)	(0.0124)	(0.0162)	(0.0198)
M_B	0.0002***	0.0006*	-0.0002	0.0000	0.0007***	0.0001
	(0.0000)	(0.0003)	(0.0004)	(0.0000)	(0.0002)	(0.0005)
CAPEXP	-0.0006***	-0.0008***	0.0001***	-0.0003***	-0.0009***	0.0002**
	(0.0001)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)
FSIZE	-0.0005	-0.0017	0.0193***	-0.0023**	0.0010	0.0190***
	(0.0010)	(0.0013)	(0.0022)	(0.0010)	(0.0013)	(0.0025)
DIV	-0.0865	-0.4850***	-0.5150***	-0.0866	-0.4550***	-0.6760***
	(0.0713)	(0.0727)	(0.1330)	(0.0740)	(0.0845)	(0.1590)
DEBT	-0.0240***	-0.0278**	0.0242***	0.0197**	-0.0187*	0.0301
	(0.0054)	(0.0134)	(0.0086)	(0.0097)	(0.0109)	(0.0189)
CFO_Vol	0.0896***	0.1570***	0.0493	0.1200***	0.1130***	0.0866***
	(0.0224)	(0.0304)	(0.0316)	(0.0085)	(0.0342)	(0.0237)
Cop_Effect	0.0136***	0.0128***	-0.0043***	0.0126***	0.0021	-0.0110
	(0.0010)	(0.0019)	(0.0015)	(0.0029)	(0.0063)	(0.0068)
DID				0.0081**	0.0154*	0.0056
				(0.0032)	(0.0087)	(0.0084)
Constant	0.0485***	0.0400	-0.3060***	0.0553***	-0.0103	-0.2800***
	(0.0150)	(0.0252)	(0.0345)	(0.0114)	(0.0242)	(0.0354)
Observations	569	528	470	569	528	458
AR(1)	0.0143	0.0022	0.0098	0.0122	0.0040	0.0059
AR(2)	0.2363	0.0754	0.0572	0.2533	0.0697	0.0196
Sargan Test	1	1	1	1	1	1

Appendix A: Variable definitions and measures

Variable	Definition	Measure
Dependent Variable		
CASH	Cash ratio	Cash divided by total assets
Ind_Adj_Cash	Industry-adjusted cash	Firm's ratio of cash to net assets and the median level of
		this ratio for the given industry
N_CASH	Net cash ratio	Cash divided by net assets
CR	Current ratio	Current assets divided by current liabilities
Independent Variable		
WDIR	Percentage of women	Ratio of total women directors to total number of
	directors	directors
BLAU_INDEX	Women on board index	$1 - \sum_{r=1}^{n} P_r^2$
		where $n$ is the number of categories of male and female directors and $Pr$ is the percentage of each category of
		male and female directors.
SHANON_INDEX	Women on board index	$-\sum_{r=1}^{n} P_r \ln P_r$
		where $n$ is the number of categories of male and female
		directors and $Pr$ is the percentage of each category of
WDID 1	W	male and female directors.
WDIR1	Women directors on	Dummy variable coded 1 if firms have at least one
WDIR2	Women directors	female on the board, 0 otherwise.  Dummy variable coded 1 if firms have at least two
WDIKE	women directors	female on the board, 0 otherwise.
WDIR3	Women directors	Dummy variable coded 1 if firms have at least three
		female on the board, 0 otherwise
BIND	Board independence	Ratio of independent directors to total number of
DCIZE	December 1	directors
BSIZE	Board size	Natural logarithm of total board members on corporate board
CEODIIAI	CEO 1 12	
CEODUAL	CEO duality	Dummy variable, coded 1 if the CEO is also chair of the
		corporate board, 0 otherwise
CFO	Operating cash flows	Pre-tax profits plus depreciation and amortization,
		divided by total assets
DEBT	Debt ratio	Total debt divided by total assets
FSIZE	Firm size	Natural logarithm of total assets
NWC	Net working capital	Current assets minus current liabilities and cash
		holdings, divided by total assets
CAPEXP	Capital expenditure	Capital expenditure to total assets
DIV	Dividends	Dividends to total assets
CF_Vol	Cash flow volatility	Standard deviation of cash flows from operations to total assets
M_B	Firms' growth	The sum of the book value of liabilities and market
		value of equity divided by the book value of total assets.

Appendix B: Measuring Excess Cash E\_Cash

To measure E\_cash, following Dittmar and Mahrt-Smith (2007), we first estimate a regression to determine the optimal cash, and excess cash is then defined as the difference between actual cash and predicted optimal cash that firms should hold. In other words, the residual of a corporate cash level regression. The main idea of the literature on optimal cash is that firms do not need to hold cash, provided that financial markets are perfect. However, due to market imperfections, cash cannot be raised instantaneously, and therefore, firms stockpile it. The equation and the definition of variables for measuring optimal cash in the regression equation are mainly based on the work of (Dittmar and Mahrt-Smith 2007) and are discussed below:

 $Ln (Cash/TNA)_{i,t} = \beta_0 + \beta_1 Ln(TNA)_{i,t} + \beta_2 (NWC/TNA)_{,t} + \beta_3 (FCF/TNA)_{i,t} + \beta_4 (R_D/TNA)_{i,t} + \beta_5 (MV/TNA)_{i,t} + \beta_6 (Industry_Vol)_{i,t} + \varepsilon_{i,t}.$ 

Where Cash is cash and equivalents, TNA is total net assets and equal to total assets minus cash and equivalents. NWC is the difference of current assets and current liabilities. The variable FCF is free cash flows; MV is market value measured as price times shares plus total liabilities. R\_D are research and development expenditures, set to zero if missing. Finally, the variable industry sigma is measured as a standard deviation of FCF of the prior 10 years.